Author’s response to reviews

Title: A TMEFF2-REGULATED CELL CYCLE DERIVED GENE SIGNATURE IS PROGNOSTIC OF RECURRENCE RISK IN PROSTATE CANCER

Authors:

Constantin Georgescu (Constantin-Georgescu@OMRF.org)
Joshua Corbin (Joshua-Corbin@ouhsc.edu)
Sandra Thibivilliers (Thibivisan@yahoo.fr)
Zachary Webb (Zachary-Webb@ouhsc.edu)
Yan Zhao (daniel-zhao@ouhsc.edu)
Jan Koster (jankoster@amc.uva.nl)
Kar-Ming Fung (KarMing-Fung@ouhsc.edu)
Adam Asch (Adam-Asch@ouhsc.edu)
Jonathan Wren (jonathan-wren@omrf.org)
Maria J Ruiz Echevarria (Maria-RuizEchevarria@ouhsc.edu)

Version: 2 Date: 05 Apr 2019

Author’s response to reviews:

AUTHORS’ RESPONSE TO REVIEWS:

Manuscript title: A TMEFF2-regulated cell cycle derived gene signature is prognostic of recurrence risk in prostate cancer

Manuscript ID: BCAN-D-18-02133

Authors:

Constantin Georgescu (Constantin-Georgescu@omrf.org)
Joshua M. Corbin (Joshua-Corbin@ouhsc.edu)
Sandra Thibivilliers (sthibi@unl.edu)
Zachary D. Webb (Zachary-Webb@ouhsc.edu)
Yan D. Zhao (Daniel-Zhao@ouhsc.edu)
Jan Koster (jankoster@amc.uva.nl)
Kar-Ming Fung (KarMing-Fung@ouhsc.edu)
Adam S. Asch (Adam-Asch@ouhsc.edu)
Jonathan D Wren (Jonathan-Wren@omrf.org)
Maria J Ruiz Echevarria (Maria-Ruizechevarria@ouhsc.edu). Corresponding author.

Please see below the details of the revisions made:

1. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

The email address of all the authors have been included in the title page. The corresponding author was indicated in the list of authors with a # symbol and an explanation to the symbol. We have now also added the words “corresponding author” following the email address of the appropriate author in the title page.

2. Please change the Introduction heading to Background.

The “Introduction” heading was changed to “Background”,

3. Thank you for uploading the sequencing data to GEO database. Since your submission will be accepted upon successful deposition of the data, please notify GEO that your submission is ready for acceptance and request that they make the data publicly available (please provide this email as guarantee of acceptance if required). In the revised version of the manuscript, please ensure to
refer to the repository in the Availability of Data and Materials section. Please provide the permanent link to the repository where the data is stored, list any deposition numbers and provide any permission required to access the data.

The data has been made publically available. It is located in the GEO repository under the series number GSE117180. We have included this information in the “declarations” section of the manuscript. In addition, we have included a link to the GEO repository, (http://www.ncbi.nlm.nih.gov/geo/), and a link directly to our data. (https://www.ncbi.nlm.nih.gov/geo/query/acc.cgi?acc=GSE117180)

4. Please refer to the authors by their initials in the Competing Interest section.

The Competing interests section has been amended. The authors are now listed by their initials.

5. Please note, the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript should be declared in the Funding section.

The funding section has been amended to include that the funding bodies have no role in the design of the study and collection, analysis, and interpretation of data or in writing the manuscript.

6. Please remove the funding information from the Acknowledgements and include it in the Funding section instead. If you have no further acknowledgements please put “Not Applicable” in the Acknowledgements section.

The funding information has been moved from the acknowledgments to the funding section.

Please note that the QC process identified two not validated references. We re-check these references and they are in pubmed as listed. We have included here the PMID numbers.